Author's response to reviews

Title: ADAR2 editing activity in newly diagnosed versus relapsed pediatric high-grade astrocytommas

Authors:

Sara Tomaselli (sara.tomaselli@opbg.net)
Federica Galeano (federicagaleano80@yahoo.it)
Luca Massimi (lucamax30@hotmail.com)
Concezio M Di Rocco (cdirocco@rm.unicatt.it)
Libero Lauriola (llibero@rm.unicatt.it)
Angela Mastronuzzi (angela.mastronuzzi@opbg.net)
Franco Locatelli (franco.locatelli@opbg.net)
Angela Gallo (angela.gallo@opbg.net)

Version: 5 Date: 5 February 2013

Author's response to reviews: see over
Dear Christina Chap,

Many thanks for your support and the acceptance of our Case Report titled "ADAR2 editing activity in newly diagnosed versus relapsed pediatric high-grade astrocytomas" (MS: 1468846291761783) for publication in *BMC Cancer* journal.

Below you will find our point-by-point.

**Point-by-point letter**

We have modified the patients’ age both in Table 1 and within the manuscript, in order to protect their anonymity as requested. In particular, we have changed the age at diagnosis as follow:

Case 1: from 12.6 years to \( \leq 13 \) years
Case 2: from 12 years to \( \leq 12 \) years
Case 3: from 7.5 years to \( \leq 8 \) years
Case 4: from 2.3 years to \( \leq 3 \) years.

We have also updated the OS of Case 4 (from 51 to 55 months) reported in Table 1.

We have formatted the new version of the manuscript conforming to the journal style.

I am looking forward to your reply.

Sincerely,

Dr. Angela Gallo

Dr. Angela Gallo
RNA Editing lab.
IRCCS OPBG
Piazza S. Onofrio, 4; 00165 Rome, Italy
Email:angela.gallo@opbg.net
Phone:+390668592659, fax: +390668592904